

Author's response to reviews

Title: Semi-automated brain tumor segmentation on multi-parametric MRI using regularized non-negative matrix factorization.

Authors:

Nicolas Sauwen (nicolas.sauwen@kuleuven.be)

Marjan Acou (marjan.acou@ugent.be)

Diana Sima (diana.sima@kuleuven.be)

Jelle Veraart (Jelle.Veraart@nyumc.org)

Frederik Maes (frederik.maes@kuleuven.be)

Uwe Himmelreich (uwe.himmelreich@kuleuven.be)

Eric Achten (Rik.Achten@ugent.be)

Sabine Van Huffel (Sabine.VanHuffel@kuleuven.be)

Version: 3 **Date:** 08 Apr 2017

Author's response to reviews:

Not applicable, only some minor editorial changes